Author's response to reviews

Title: Lactobacillus casei strain GG in the treatment of infants with acute watery diarrhea: A randomized, double-blind, placebo controlled clinical trial

Authors:

Eduardo Salazar-Lindo (edsalaza@ec-red.com)
Percy Miranda-Langschwager (percy_miranda@bonus.com.pe)
Miguel Campos-Sanchez (viper@upch.edu.pe)
Elsa Chea-Woo (echea@upch.edu.pe)
Richard B Sack (rsack@jhsph.edu)

Version: 3 Date: 1 September 2004

Author's response to reviews:

August 31, 2004

Ms. Emma Parkin
Editorial Administrator
The BioMed Central Editorial Team

Re: "Lactobacillus casei strain GG in the treatment of infants with acute watery diarrhea: A randomized, double-blind, placebo controlled clinical trial"

Dear Ms. Parkin,

I'm pleased to submit the final version of our manuscript for your consideration. We have made the following changes in the manuscript:

1. ISRCTN has been incorporated at the end of the title.
2. Author's qualifications have been removed from the title page.
3. All tracking has been turned off.
4. References in the text are now in square brackets and not superscript.
5. The phrase "The authors declare that they have no competing interests" has been added to the Competing Interests section.
6. The text ON A SEPARATE FILE has been removed and replaced by "Figure 1" in the figure legend.
7. All tables are now portrait oriented.
8. White space around the figure has been reduced.

I look forward to your kind response.

Yours sincerely,

Eduardo Salazar-Lindo, M.D.
Professor and Chair
Department of Pediatrics
Cayetano Heredia Hospital
Av. El Polo 740 Of. C-410
Surco, Lima 33, PERU